Author's response to reviews

Title: Using Normalisation Process Theory to Design Translational Research: A review of a speech and language therapy intervention

Authors:

Deborah M James (deborah.james@nottingham.ac.uk)

Version: 2 Date: 29 June 2010

Author's response to reviews:

I have conducted revisions as requested:

included the title page in the main document and included the acknowledgements in the main document
edited the reference list so that all authors are listed for all publications cited
taken the references out of the abstract
included text that was stand alone as an appendix in the main body of the test